Author’s response to reviews

Title: Barriers to asymptomatic screening and other STD services for adolescents and young adults: focus group discussions

Authors:

Dr Elizabeth C Tilson (elizabeth.tilson@co.wake.nc.us)
Victoria Sanchez (vsanchez@email.unc.edu)
Chandra L Ford (clford@email.unc.edu)
Marlene Smurzynski (msmurzyn@email.unc.edu)
Peter A Leone (pal007@med.unc.edu)
Kimberley K Fox (kaf6@cdc.gov)
Kathleen Irwin (kli1@cdc.gov)
Dr William C Miller (Bill_Miller@unc.edu)

Version: 5 Date: 8 Jun 2004

Editors, BMC Public Health

Thank you very much for accepting our manuscript for publication. The following requested format changes have been made.

Title page:
All unnecessary capitalisation have been removed from both the manuscript file and the submission system.
The authors' qualifications have been removed from the title page.
The country has been included in the affiliation details.
The "support" section has been moved from the title page to the "Acknowledgements" section.

Abstract:
The word count has been removed.
The keywords have been incorporated in the title and abstract and the "keyword" section has been removed.

Manuscript:
References are cited in the text using consecutive numbers in square brackets.

Tables:
The borders of tables include visible black lines and the tables are divided into cells / fields.
Each separate quote in the tables are in a separate cell.
Each table starts on a new page.
Tables 2 & 3 - Each table has been made into just one table, that runs over to the next page, with the repeated set of column headings removed.

I hope you find these changes acceptable and thank you again.

Elizabeth Tilson